Author’s response to reviews

Title: Prevalence and outcomes of breast milk expressing in women with healthy term infants: a review of the literature

Authors:

Helene M Johns (hjohns@students.latrobe.edu.au)
Della A Forster (D.Forster@latrobe.edu.au)
Lisa H Amir (L.Amir@latrobe.edu.au)
Helen L McLachlan (H.McLachlan@latrobe.edu.au)

Version: 3 Date: 9 October 2013

Author’s response to reviews:

Ms Helene Johns
Mother & Child Health Research
La Trobe University
215 Franklin St
Melbourne VIC 3000
Australia

9 October 2013

The Editor
BMC Pregnancy and Childbirth

Dear Ms Cruz,

Thank you for your correspondence notifying acceptance of our manuscript entitled “Prevalence and outcomes of breast milk expressing in women with healthy term infants: a review of the literature.”

We would like the tables to be visible in the final published manuscript and to this end have included them in a tables section following the references.

Yours sincerely,

Helene Johns